## RESEARCH





# Intimate partner violence victimization during pregnancy increases risk of postpartum depression among urban adolescent mothers in South Africa

Luwam T. Gebrekristos<sup>1\*</sup>, Allison K. Groves<sup>2</sup>, Luz McNaughton Reyes<sup>3</sup>, Dhayendre Moodley<sup>4</sup>, Mags Beksinska<sup>5</sup> and Suzanne Maman<sup>3</sup>

## Abstract

**Background** It is estimated that 38.8% of mothers develop postpartum depression (PPD) in South Africa. While empirical evidence documents an association between intimate partner violence (IPV) victimization in pregnancy and PPD among adult women, the association has been underexamined among adolescent mothers (< 19 years). The study's purpose is to examine whether IPV victimization during pregnancy is associated with PPD among adolescent mothers.

**Methods** Adolescent mothers (14–19 years) were recruited at a regional hospital's maternity ward in KwaZulu Natal, South Africa between July 2017-April 2018. Participants completed behavioral assessments at two visits (n = 90): baseline (up to 4 weeks postpartum) and follow-up (6–9 weeks postpartum, when PPD is typically assessed). The WHO modified conflict tactics scale was used to create a binary measure of any physical and/or psychological IPV victimization that occurred during pregnancy. Participants with scores  $\geq$  13 on the Edinburgh Postpartum Depression Scale (EPDS) were classified as having symptoms of PPD. We used a modified Poisson regression with robust standard errors to assess PPD in association with IPV victimization during pregnancy, controlling for relevant covariates.

**Results** Nearly one-half (47%) of adolescent mothers reported symptoms of PPD by 6–9 weeks post-delivery. Further, IPV victimization during pregnancy was highly prevalent (40%). Adolescent mothers who reported IPV victimization during pregnancy had marginally higher risk of PPD at follow-up (RR: 1.50, 95 CI: 0.97-2.31; p = 0.07). The association was strengthened and significant in covariate-adjusted analysis (RR: 1.62, 95 CI: 1.06-2.49; p = 0.03).

**Conclusions** Poor mental health was common among adolescent mothers, and IPV victimization during pregnancy was associated with PPD risk among adolescent mothers. Implementing IPV and PPD routine screenings during the perinatal period may aid in identifying adolescent mothers for IPV and PPD interventions and treatment. With the high prevalence of IPV and PPD in this vulnerable population and the potential negative impact on maternal and infant outcomes, interventions to reduce IPV and PPD are needed to improve adolescent mothers' well-being and their baby's health.

\*Correspondence: Luwam T. Gebrekristos Ig526@drexel.edu Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

**Keywords** Adolescent mothers, Postpartum depression, Intimate partner violence, Pregnancy, South Africa, Maternal health

## **Plain language summary**

**Background** More than one-third of adult mothers experience postpartum depression (PPD) in South Africa and intimate partner violence (IPV) victimization is a strong risk factor of PPD for adult mothers. However, there are no studies on adolescent mothers that look at the link between IPV victimization and PPD. This paper aims to examine whether IPV victimization during pregnancy is associated with PPD among adolescent South African mothers.

**Methods** We had 90 adolescent mothers (aged 14–19 years old) complete an initial survey between delivery and 4 weeks postpartum to collect information on IPV during their pregnancy. Participants completed an additional survey between 6 and 9 week postpartum to collect information on the symptoms of PPD.

**Results** Nearly one-half (47%) of adolescent mothers reported symptoms of PPD by 6–9 weeks post-delivery. Report of IPV victimization during pregnancy was also very high (40%). Adolescent mothers who experienced IPV victimization during pregnancy were more likely to report symptoms of PPD between 6 and 9 weeks postpartum.

**Conclusions** PPD and IPV was very common in our sample, and IPV victimization during pregnancy was linked to PPD among adolescent mothers. Having routine screenings during pregnancy and postpartum period can identify adolescent mothers IPV and PPD interventions and treatment. With the high reports of IPV and PPD in this sample and the potential negative impact on maternal and infant outcomes, interventions to reduce IPV and PPD are needed to improve adolescent mothers' well-being and their baby's health.

## Background

The estimated global postpartum depression (PPD) prevalence is 17.2% (95% CI: 16.0–18.5) and higher in lowand middle-income countries [1, 2]. Existing research has linked PPD to numerous negative infant-related outcomes: impaired cognitive or motor development [3–5], low-growth trajectory [3, 6–8], and behavioral inhibition and delayed emotional development [4, 9–11]. Further, PPD negatively impacts maternal social wellbeing (e.g., homelessness, anxiety, substance abuse and suicidal ideation) [12–14] and mother-infant bonding [15, 16].

The prevalence of PPD among adult mothers in South Africa is more than double the global rate (38.8%; 95% CI: 25.7%–53.7%) [1]. Despite the negative infant and maternal consequences of PPD, it is underdiagnosed and routine screening is not a part of standard postpartum care in South Africa [17, 18]. Extant literature documents prior history of depressive symptoms [19–21], unwanted pregnancy [19, 22, 23], being unmarried [19], intimate partner violence (IPV) victimization [22, 24], socioeconomic vulnerabilities [24, 25], and low social support [19–25] as risk factors for PPD in South Africa.

While these studies highlight the prevalence and risk factors of PPD among South African adult mothers, there are gaps in knowledge of the prevalence of PPD and risk factors of PPD for adolescent mothers. One cross-sectional study limited to adolescent mothers (aged 14–19) estimated prevalence of PPD at 8.8% [26]. However, PPD

was measured at 1 week postpartum, which is not the standard timing for PPD assessment (4–6 weeks postpartum) [27]. The same study identified IPV victimization, verbal abuse, and physical violence as risk factors of PPD among adolescent mothers [26]. However, with the crosssectional study design, establishing whether violence was a cause or consequence of PPD is not possible.

Adolescent mothers are largely excluded from studies on postpartum mental health; and yet, the unique circumstances associated with adolescent pregnancy in South Africa may heighten adolescent mothers' risk of PPD. First, structural factors in post-apartheid South Africa impact the mental health of South Africans, particularly those who are marginalized like black adolescent mothers [28–31]. Legacies of apartheid, like material inequalities, are common among Black adolescent mothers and are risk factors of poor mental health [20, 32, 33]. Second, in two qualitative studies, South African adolescent mothers describe how lack of support and isolation by partners, families and communities during pregnancy led them to experience poor mental health postpartum [34, 35]. At the same time, adolescent mothers are more likely to test positive for HIV during the perinatal period than adult mothers in sub-Saharan Africa [36], which can result in depressive symptoms [37]. Finally, adolescent mothers can experience social stigma which may inhibit engagement in postpartum clinic-based care [38, 39].

Poor engagement in postpartum care may lead to large diagnosis and treatment gaps for adolescent mothers.

Adolescent mothers are usually excluded from studies, yet the unique circumstances,- including the way their intimate relationships are configured - may heighten their risk of PPD. Specifically, several studies demonstrate that adolescent mothers are more likely to have risk factors for PPD compared to adult mothers in that they are more likely to: experience IPV victimization, have relationships with older men (who may be more likely to perpetrate IPV than peer-aged partners) [40], be unmarried [41], and have an unintended pregnancy [41]. These risk factors of PPD are either relationship characteristics or can negatively impact power in their relationships, which highlights the importance of examining relationship characteristics in understanding PPD among adolescent mothers. For example, in population-based studies across several countries women who had an unintended pregnancy were more likely to experience IPV victimization than those who had an intended pregnancy [42].

Some adolescent mothers fare better on risk factors of PPD than other adolescent mothers, which emphasizes the need to conduct research in this population. In South Africa, where the adolescent birth rate is 1.5 times more the global rate [43], risk factors for PPD like unintended pregnancies and being unmarried are extremely common among adolescent mothers [41, 44]; however, there is evidence of variability in IPV victimization. Despite this, no longitudinal studies have examined the link between IPV victimization during pregnancy and symptoms of PPD among adolescent mothers. Understanding the burden of PPD and the link between IPV victimization and symptoms of PPD among this population is needed to identify which adolescent mothers are most at-risk. Thus, the objective of this study is to test the hypothesis that experiencing IPV victimization during pregnancy increases the likelihood of reporting symptoms of PPD among adolescent mothers. This study addresses the current gaps in the literature by leveraging longitudinal data to estimate the prevalence of PPD at 6-9 postpartum and examine the association between IPV during pregnancy and PPD among adolescent mothers in South Africa.

## Methods

## Study design

Data for the analyses comes from a pilot study, Mentoring Adolescent Mothers at School (MAMAS). The MAMAS study was designed to support adolescent mothers return to school following delivery [45]. Screening for eligibility occurred between July 2017 and April 2018 at the regional hospital's maternity ward in KwaZulu Natal, South Africa. Mothers were eligible to participate in MAMAS if they had attended school in the past year, were between 14 and 19 years old, neither they nor their baby had any major health issues and had planned to live in the township for 6 months (n = 119).

Eligible and interested participants (and parents/guardians of minors) provided informed consent and/or assent prior to initiation of study activities [46]. All enrolled participants completed a STI testing and behavioral survey at their baseline visit (1 day to 4 weeks postpartum) and follow-up visit (between 6–9 weeks postpartum). Participants in the intervention arm were invited to participate in a pilot intervention which consisted of up to 10 group mentoring sessions facilitated by a mentor mother (a young adult mother who had been pregnant as an adolescent and completed high school after delivery). Sessions were delivered from 6 weeks to 6 months postpartum. All study activities were approved by the institutional ethic boards at Drexel University (1612005048) and University of KwaZulu Natal (BFC023/17).

## Measures

The dependent variable in this study was whether the participant reported symptoms of PPD in the past seven days. Participants completed the 10-item Edinburgh Postnatal Depression Scale (EPDS) between 6 and 9 weeks postpartum to identify those with depressive symptoms in the past week [47]. EPDS items were scored on a four-point Likert scale with higher scores indicating concerning symptoms. Scores for the 10 items were summed to generate a final score between 0 and 30 ( $\alpha$ =0.75). A cutoff score of  $\geq$  13 has been used to indicate depressive symptoms; [47, 48] participants with scores  $\geq$  13 were coded as having symptoms of PPD. Those with scores <13 were coded as having no symptoms of PPD.

The independent variable, IPV victimization during pregnancy, was measured retrospectively at baseline (no more than 4 weeks postpartum) using 10 items from the WHO modified conflict tactics scale on physical and psychological violence [49]. Participants reported the frequency at which they experienced each of the 10 acts of violence (e.g., during your pregnancy, has a romantic or sexual partner ever push or shoved you). Adolescent mothers who reported at least 1 act of physical or psychological violence were coded as experiencing IPV victimization during pregnancy. Those who did not report any acts of physical or psychological violence were coded as no IPV victimization during pregnancy.

We controlled for baseline covariates that we hypothesize would be associated with the independent and dependent variables: age (in years) [41, 50–52], socioeconomic status (SES) [25, 53], HIV status [54–56], STI diagnosis [57, 58], perceived social support [20, 25, 26, 59], number of children [60, 61], and symptoms of PPD [20, 26]. We created an index for baseline SES [62, 63], composed of 18 items of working household facilities (e.g., hot running water, electricity, refrigerator). A principal component analysis of the 18 items loaded into a single principal component. Scores lower than the 40th percentile were coded as low SES, scores between 40 and 80th percentile were coded as middle SES, and scores above the 80th percentile were coded as high SES. In South Africa, HIV testing occurs during antenatal care visits and delivery [64]. Participants' HIV status was determined by reviewal of maternity health records. During the baseline visit, participants completed STI testing for 3 pathogens: Neiserria gonorrhea (Gonorrhea), Chlamydia trachomatis (Chlamydia), and Trichomonas vaginalis (Trichomonas). Participants who tested positive for at least 1 pathogen were coded as having an STI diagnosis. Participants who tested negative on all 3 pathogens were coded as not having an STI diagnosis. Perceived social support was measured using 12-items to assess support from family, friends, and partner (e.g., My family really tries to help me; I can talk about my problems with my friends; I have a romantic partner who is a real source of comfort to me). Items were scored on a 4-point Likert scale with higher scores indicating higher social support. Number of children at the time of their delivery was determined by reviewing maternity health records: 1 child versus 2 children (none of the participants had more than 2 children). Symptoms of PPD within 4 weeks postpartum, like the dependent variable, was measured using EPDS (scores  $\geq$  13 versus scores < 13,  $\alpha$  = 0.82). Finally, we also controlled for time between baseline and follow-up visits (in weeks) and intervention attendance (attended at least one session versus attended zero sessions).

#### Data Analysis

We excluded participants who did not complete the follow-up visit (n=25). We further excluded participants who were missing covariates or the outcome (n=4), yielding a total sample size of 90 participants. Neither IPV during pregnancy nor baseline symptoms of PPD were associated with study dropout. For our analyses, we used bivariate and multivariable modified Poisson regression with robust standard errors (generates risk ratios) to assess whether IPV victimization during pregnancy is associated PPD, controlling for covariates ( $\alpha$ =0.05) [65, 66]. All analyses were conducted using SAS Version 9.4. [67]

#### **Table 1** Characteristics of study participants (N = 90)

	All participants N=90 n(%) or mean (sd)
Dependent variable	
Symptoms of PPD at follow-up	42 (46.7%)
Independent variable	
IPV during pregnancy	36 (40.0%)
Baseline covariates	
Age (in years)	17.5 (1.4)
SES	
Low	37 (41.1%)
Middle	35 (38.9%)
High	18 (20.0%)
Positive HIV status	21 (23.3%)
STI diagnosis	16 (17.8%)
Perceived Social Support Score	3.2 (0.5)
Two children at time of delivery	10 (11.1%)
Symptoms of PPD within 4 weeks postpartum	59 (65.6%)
Attended at least 1 intervention session	39 (43.3%)
Timing between baseline and follow-up (in weeks)	5.4 (1.4)

## Results

Nearly half of adolescent mothers reported symptoms of PPD between 6 and 9 weeks postpartum (Table 1; 46.7%) and approximately two-thirds had symptoms of PPD at 4 weeks postpartum or less (65.6%). More than one-third of adolescent mothers reported experiencing IPV during pregnancy (40.0%), with more than onethird reporting any psychological violence (36.7%) and nearly one in five adolescent mothers reporting physical violence (16.7%). Nearly one quarter of participants were HIV-positive (23.3%) and 17.8% of participants had an STI diagnosis at baseline.

Table 2 illustrates bivariate associations between IPV, baseline covariates and PPD (Model 1). IPV during pregnancy was marginally associated with symptoms of PPD at follow-up (risk ratio (RR): 1.50; 95% CI: 0.97, 2.31). The risk of PPD symptoms at follow-up was more than double among adolescent mothers who reported symptoms of PPD within 4 weeks postpartum than those who did not report symptoms of PPD within 4 weeks postpartum (RR: 2.23; 95% CI: 1.18, 4.22). No other baseline covariates were associated with PPD at follow up.

Table 2 also shows multivariable associations (Model 2). The association between IPV during pregnancy and symptoms of PPD at follow-up was strengthened when controlling for baseline covariates. When accounting for baseline covariates, adolescent mothers who experienced

	Model 1 <sup>a</sup>		Model 2 <sup>b</sup>	
	RR (95% CI)	p-value	RR (95% CI)	p-value
IPV during pregnancy	1.50 (0.97, 2.31)	0.07	1.62 (1.06, 2.49)	0.03
Age (in years)	1.06 (0.89, 1.27)	0.51	0.99 (0.82, 1.20)	0.95
SES				
Low	Ref.		Ref.	
Middle	1.12 (0.68, 1.86)	0.65	1.04 (0.66, 1.64)	0.86
High	1.16 (0.64, 2.09)	0.63	1.13 (0.60, 2.14)	0.71
HIV-positive	1.31 (0.83, 2.08)	0.24	1.33 (0.82, 2.18)	0.25
STI diagnosis	0.77 (0.39, 1.51)	0.45	1.00 (0.54, 1.85)	0.99
Perceived Social Support Score	0.92 (0.62, 1.37)	0.69	0.93 (0.60, 1.44)	0.75
Two children at time of delivery	1.08 (0.56, 2.10)	0.82	0.96 (0.50, 1.83)	0.9
Symptoms of PPD within 4 weeks postpartum	2.23 (1.18, 4.22)	0.01	2.49 (1.31, 4.72)	0.01
Attended at least 1 intervention session	0.89 (0.56, 1.40)	0.61	0.82 (0.49, 1.38)	0.46
Timing between baseline and follow-up (in weeks)	0.98 (0.82, 1.15)	0.77	0.91 (0.78, 1.07)	0.27

 Table 2 Bivariate and Multivariable Associations between IPV during pregnancy, baseline covariates and symptoms of PPD at follow-up

IPV intimate partner violence

PPD postpartum depression

RR risk ratios

<sup>a</sup> Model 1: Bivariate associations

<sup>b</sup> Model 2: Multivariable associations

IPV during pregnancy were 1.62 times as likely to report symptoms of PPD at follow-up than those who did not experience IPV during pregnancy (95% CI: 1.06, 2.49). Further, adolescent mothers who had symptoms of PPD within 4 weeks postpartum were 2.49 times as likely to report symptoms of PPD at follow-up compared to those who did not have symptoms of PPD within 4 weeks postpartum (95% CI: 1.31, 4.72) when controlling for other variables.

## Discussion

Despite the high adolescent birth rate and unique context of adolescent pregnancy, the association between IPV and PPD has been understudied among adolescent mothers in South Africa. The purpose of the study was to measure the prevalence of PPD symptoms and examine the link between IPV during pregnancy and symptoms of PPD among adolescent mothers in South Africa. In this study, almost half of adolescent reported symptoms of PPD between 6 and 9 weeks postpartum. Further, experiencing IPV during pregnancy was associated with an increased likelihood of reporting symptoms of PPD.

Studies on PPD among adolescent mothers in sub-Saharan Africa do not measure PPD consistently. Specifically, in two studies, time since delivery varied among study participants and another study measured PPD at 1 week postpartum [26, 68, 69]. Given the variability in PPD assessment within and across studies, it is difficult to compare our study findings to existing studies with adolescent mothers. Nonetheless, our prevalence estimate was higher than depression prevalence estimates among South African adolescent girls (7.3–35.9%) [70– 73] and was on the higher end of PPD prevalence estimates among adult mothers (16.7–49.3%). [17, 20–22, 74]

Consistent with our hypothesis, experiencing IPV victimization during pregnancy was associated with the increased likelihood of reporting symptoms of PPD between 6–9 weeks in covariate-adjusted analysis. However, bivariate analysis for IPV victimization was only marginally significant. The reason for statistically significant association in the covariate-adjusted analysis is unclear but may be attributed to negative confounding. Nonetheless, our findings highlight the pervasiveness of IPV victimization among this population and the need for adolescent-friendly IPV screenings and interventions in maternal health care.

In addition to experiencing IPV victimization during pregnancy, reporting symptoms of PPD within 4 weeks postpartum was also associated with increased likelihood of reporting symptoms of PPD between 6 and 9 weeks. Studies in sub-Saharan Africa on adult and adolescent mothers' mental health have identified history of poor mental health as a predictor of poor postpartum mental health [19, 75, 76]. Further, nearly two-thirds of our sample reported symptoms of PPD within 4 weeks postpartum. Though depressive symptoms in the first month postpartum may indicate baby blues instead of PPD, it was our strongest predictor of PPD between 6 and 9 weeks postpartum. Our findings-along with the broader empirical literature-suggest that we might consider repeated screenings for poor maternal mental health more throughout the perinatal period. Recognizing that there is limited research on postpartum mental health among adolescent mothers, there are several opportunities for future research. While the prevalence of IPV victimization is high, IPV experiences and relationship dynamics are not monolithic in this population. Thus, future research is needed to understand which adolescent mothers are most at-risk for PPD by examining other salient relationship factors beyond IPV, like partner support. Further research is also needed to identify protective factors that mitigate adolescents' risk of violence during pregnancy and poor mental health postpartum. Finally, given the South African context, further research on structural factors (e.g., income inequality) impact on PPD among adolescent mothers is needed.

In addition, there are several policy and intervention implications. Specifically, clinical providers should implement screenings for IPV during antenatal care visits and screenings for symptoms of PPD during postpartum visits. Despite WHO recommendations [77, 78], routine screenings for PPD and IPV are not included in guidelines for standard of postpartum care in South Africa [18]. Recognizing that optimally-timed screenings is a starting point for additional care rather than an end point, screening without referrals for psychosocial interventions or treatment is not recommended [77]. Acknowledging that adolescent mothers may face stigma by healthcare workers, a barrier to engaging in care [38, 39], providing adolescent-friendly maternal care may increase access to mental health care. Given the lack of interventions that address the specific needs of adolescent mothers, further attention is needed to develop IPV and PPD interventions for adolescent mothers. With the detrimental impact of PPD on both maternal and infant outcomes, it is vital to include adolescent mothers in postpartum mental health research to minimize the burden of IPV victimization and PPD in this vulnerable population.

Our study has several limitations. First, our small sample size precludes our ability to examine physical and psychological IPV victimization separately. It is possible that physical and psychological IPV victimization impact PPD differently. Relatedly, we could not examine sexual IPV victimization. Second, our independent variable, symptoms of PPD, is self-reported not a clinical diagnosis. Although the self-report measure does not replace a clinical diagnosis, studies have shown that EPDS is a strong predictor for clinical diagnosis of PPD [79, 80]. Finally, adolescent mothers were ineligible to participate in the parent study if they reported poor maternal or birth outcomes. Therefore, our findings may not be generalizable to adolescent mothers with poor maternal or birth outcomes. Considering that poor birth outcomes are associated with symptoms of PPD in adult mothers [25], we expect the prevalence of PPD symptoms to be higher in the excluded population.

Nonetheless, this study has many strengths. Our study highlights an overlooked population in postpartum mental health research—adolescent mothers. Further, this study is the first to examine the relationship between IPV during pregnancy and PPD among adolescent mothers in South Africa. Further, we used longitudinal design. Finally, we used a validated and WHO-recommended screener, EPDS, to measure our outcome.

## Conclusions

The burden of PPD and adolescent birth rate is elevated in South Africa. However, adolescent mothers are excluded from postpartum mental health research. Given the unique and difficult circumstances pregnant adolescent girls and adolescent mothers face and the high burden of teen pregnancy in South Africa [81, 82], it is crucial to examine adolescent postpartum mental health. This study broadens our understanding by estimating prevalence of PPD and the impact of IPV victimization on PPD. Our findings show that nearly half of adolescent mothers reported symptoms of PPD and that IPV victimization during pregnancy is associated with increased likelihood of reporting symptoms of PPD. Given the high prevalence of IPV and PPD in this vulnerable population and the negative impact on maternal and infant outcomes, interventions to reduce IPV and PPD are needed to improve adolescent mothers' well-being and their baby's health.

#### Abbreviations

EPDS	Edinburgh Postnatal Depression Scale
IPV	Intimate partner violence
MAMAS	Mentoring Adolescent Mothers at School
PPD	Postpartum depression
SES	Socioeconomic status
STI	Sexually transmitted infections

#### Acknowledgements

First, we sincerely thank the adolescent mothers who participated in our parent study. Second, we are very grateful for our research staff who collected the data.

#### Author contributions

LTG conceptualized the study, led the analysis, and developed the initial draft of the manuscript. AKG, LMR, DM, and SM designed and implemented the parent study. LTG, AKG, LMR, DM, MB, and SM contributed to interpreting the

results and editing the manuscript, reviewed, and approved the manuscript submission. All authors read and approved the final manuscript.

#### Funding

The MAMAS intervention was funded by the United States Department of State (S-LMAQM-16-CA1103; Dr. Allison K. Groves, Principal Investigator) as part of the DREAMS Innovation Challenge and managed by JSI Research & Training Institute, Inc. (JSI). The opinions and conclusions stated here are those of the authors and do not necessarily reflect those of the United States Department of State or JSI. This work was supported by National Institute on Minority Health and Health Disparities (NIMHD #1T37MD014251).

## Availability of data and materials

Data used for analyses are available from the corresponding author with reasonable, written request.

## Declarations

## Ethics approval and consent to participate

Ethics approval was obtained from the institutional ethic boards at Drexel University (1612005048) and University of KwaZulu Natal (BFC023/17). We solicited written informed consent from adolescent mothers aged 18 or older. For an adolescent mother younger than 18, we obtained written informed consent from her parent/guardian (with adolescent mother's permission) and obtained assent from the adolescent mother. All eligible adolescent mothers (and their parent/guardian, if adolescent < 18 years old) were informed of the benefits and risks of the study prior to soliciting informed consent of assent.

#### **Consent for publication**

Not applicable.

#### Competing interests

The authors declare that they have no competing interests.

#### Author details

<sup>1</sup>Department of Epidemiology and Biostatistics, Drexel University Dornsife School of Public Health, 3215 Market Street, Philadelphia, PA 19140, USA. <sup>2</sup>Department of Community Health and Prevention, Drexel University Dornsife School of Public Health, Philadelphia, PA, USA. <sup>3</sup>Department of Health Behavior, University of North Carolina at Chapel Hill Gillings School of Global Public Health, Chapel Hill, NC, USA. <sup>4</sup>Department of Obstetrics and Gynaecology, University of KwaZulu-Natal Nelson R. Mandela School of Medicine, Durban, South Africa. <sup>5</sup>MatCH Research Unit (MRU), Faculty of Health Sciences, University of the Witwatersrand, Durban, South Africa.

## Received: 1 October 2022 Accepted: 29 March 2023 Published online: 02 May 2023

#### References

- Wang Z, Liu J, Shuai H, Cai Z, Fu X, Liu Y, et al. Mapping global prevalence of depression among postpartum women. Transl Psychiatry. 2021;11(1):1–13.
- Fisher J, de Mello MC, Patel V, Rahman A, Tran T, Holton S, et al. Prevalence and determinants of common perinatal mental disorders in women in low-and lower-middle-income countries: a systematic review. Bull World Health Organ. 2012;90:139–49.
- Nasreen HE, Kabir ZN, Forsell Y, Edhborg M. Impact of maternal depressive symptoms and infant temperament on early infant growth and motor development: results from a population based study in Bangladesh. J Affect Disord. 2013;146(2):254–61.
- Ali NS, Mahmud S, Khan A, Ali BS. Impact of postpartum anxiety and depression on child's mental development from two peri-urban communities of Karachi, Pakistan: a quasi-experimental study. BMC Psychiatry. 2013;13(1):274.
- Koutra K, Chatzi L, Bagkeris M, Vassilaki M, Bitsios P, Kogevinas M. Antenatal and postnatal maternal mental health as determinants of infant neurodevelopment at 18 months of age in a mother–child cohort

- Gress-Smith JL, Luecken LJ, Lemery-Chalfant K, Howe R. Postpartum depression prevalence and impact on infant health, weight, and sleep in low-income and ethnic minority women and infants. Matern Child Health J. 2012;16(4):887–93.
- Avan B, Richter LM, Ramchandani PG, Norris SA, Stein A. Maternal postnatal depression and children's growth and behaviour during the early years of life: exploring the interaction between physical and mental health. Arch Dis Child. 2010;95(9):690–5.
- Adewuya AO, Ola BO, Aloba OO, Mapayi BM, Okeniyi JAO. Impact of postnatal depression on infants' growth in Nigeria. J Affect Disord. 2008;108(1):191–3.
- Feldman R, Granat ADI, Pariente C, Kanety H, Kuint J, Gilboa-Schechtman E. Maternal depression and anxiety across the postpartum year and infant social engagement, fear regulation, and stress reactivity. J Am Acad Child Adolesc Psychiatry. 2009;48(9):919–27.
- Moehler E, Kagan J, Parzer P, Brunner R, Reck C, Wiebel A, et al. Childhood behavioral inhibition and maternal symptoms of depression. Psychopathology. 2007;40(6):446–52.
- Conroy S, Pariante CM, Marks MN, Davies HA, Farrelly S, Schacht R, et al. Maternal psychopathology and infant development at 18 months: the impact of maternal personality disorder and depression. J Am Acad Child Adolesc Psychiatry. 2012;51(1):51–61.
- Chapman SLC, Wu LT. Postpartum substance use and depressive symptoms: a review. Women Health. 2013;53(5):479–503.
- Vliegen N, Casalin S, Luyten P, Docx R, Lenaerts M, Tang E, et al. Hospitalization-based treatment for postpartum depressed mothers and their babies: rationale, principles, and preliminary follow-up data. Psychiatry. 2013;76(2):150–68.
- Curtis MA, Corman H, Noonan K, Reichman NE. Maternal Depression as a Risk Factor for Family Homelessness. Am J Public Health. 2014;104(9):1664–70.
- Dubber S, Reck C, Müller M, Gawlik S. Postpartum bonding: the role of perinatal depression, anxiety and maternal–fetal bonding during pregnancy. Arch Women's Ment Health. 2015;18(2):187–95.
- Edhborg M, Nasreen HE, Kabir ZN. Impact of postpartum depressive and anxiety symptoms on mothers' emotional tie to their infants 2–3 months postpartum: a population-based study from rural Bangladesh. Arch Women's Ment Health. 2011;14(4):307–16.
- Hung KJ, Tomlinson M, le Roux IM, Dewing S, Chopra M, Tsai AC. Community-based prenatal screening for postpartum depression in a South African township. Int J Gynecol Obstet. 2014;126(1):74–7.
- South Africa National Department of Health. Ideal Clinic Monitoring System [Internet]. 2021 https://www.idealhealthfacility.org.za/. Accessed 30 Sep 2022.
- Abrahams JM, Stellenberg EL. Prevalence of and factors influencing postnatal depression in a rural community in South Africa. Afr J Phys Health Educ Recreat Dance. 2015;7(1):1–8.
- Ramchandani PG, Richter LM, Stein A, Norris SA. Predictors of postnatal depression in an urban South African cohort. J Affect Disord. 2009;113(3):279–84.
- Pingo J, van den Heuvel LL, Vythylingum B, Seedat S. Probable postpartum hypomania and depression in a South African cohort. Arch Womens Ment Health. 2017;20(3):427–37.
- 22. Mokwena K, Shiba D. Prevalence of postnatal depression symptoms in a primary health care clinic in Pretoria South Africa of health care services. Afr J Phys Health Educ Recreat Dance. 2014;20:116–27.
- Tomlinson M, Swartz L, Cooper PJ, Molteno C. Social Factors and Postpartum Depression in Khayelitsha, Cape Town. South Afr J Psychol. 2004;34(3):409–20.
- 24. Garman EC, Schneider M, Lund C. Perinatal depressive symptoms among low-income South African women at risk of depression: trajectories and predictors. BMC Pregnancy Childbirth. 2019;19(1):202.
- Mokwena K, Masike I. The need for universal screening for postnatal depression in South Africa: confirmation from a Sub-District in Pretoria, South Africa. Int J Environ Res Public Health. 2020;17(19):6980.
- 26. Govender D, Naidoo S, Taylor M. Antenatal and postpartum depression: prevalence and associated risk factors among adolescents' in KwaZulu-Natal, South Africa. Depress Res Treat. 2020;21(2020): e5364521.

- Sit DK, Wisner KL. The identification of postpartum depression. Clin Obstet Gynecol. 2009;52(3):456–68.
- Das-Munshi J, Lund C, Mathews C, Clark C, Rothon C, Stansfeld S. Mental health inequalities in adolescents growing up in post-apartheid South Africa: cross-sectional survey, SHaW study. PLoS ONE. 2016;11(5): e0154478.
- Burns JK, Tomita A, Lund C. Income inequality widens the existing income-related disparity in depression risk in post-apartheid South Africa: evidence from a nationally representative panel study. Health Place. 2017;45:10–6.
- Barbarin OA, Richter LM. Mandela's children: growing up in post-apartheid South Africa. Routledge; 2013. https://doi.org/10.4324/97813 15022956
- Harriman NW, Williams DR, Morgan JW, Sewpaul R, Manyaapelo T, Sifunda S, et al. Racial disparities in psychological distress in post-apartheid South Africa: results from the SANHANES-1 survey. Soc Psychiatry Psychiatr Epidemiol. 2021. https://doi.org/10.1007/ s00127-021-02175-w.
- Jewkes R, Vundule C, Maforah F, Jordaan E. Relationship dynamics and teenage pregnancy in South Africa. Soc Sci Med. 2001;52(5):733–44.
- Vundule C, Maforah F, Jewkes R, Jordaan E. Risk factors for teenage pregnancy among sexually active black adolescents in Cape Town. S Afr Med J. 2001;91(1):73–80.
- 34. Govender D, Naidoo S, Taylor M. "I have to provide for another life emotionally, physically and financially": understanding pregnancy, motherhood and the future aspirations of adolescent mothers in KwaZulu-Natal South, Africa. BMC Pregnancy Childbirth. 2020;20(1):620.
- Naidoo J, Muthukrishna N, Nkabinde R. The journey into motherhood and schooling: narratives of teenage mothers in the South African context. Int J Incl Educ. 2021;25(10):1125–39.
- Graybill LA, Kasaro M, Freeborn K, Walker JS, Poole C, Powers KA, et al. Incident HIV among pregnant and breastfeeding women in sub-Saharan Africa: a systematic review and meta-analysis. AIDS. 2020;34(5):761–76.
- Kiene SM, Dove M, Wanyenze RK. Depressive symptoms, disclosure, HIVrelated stigma, and coping following HIV testing among outpatients in uganda: a daily process analysis. AIDS Behav. 2018;22(5):1639–51.
- Govender D, Taylor M, Naidoo S. Adolescent pregnancy and parenting: perceptions of healthcare providers. J Multidiscip Healthc. 2020. https:// doi.org/10.2147/JMDH.S258576.
- Naidoo S, Taylor M, Govender D. Nurses' perception of the multidisciplinary team approach of care for adolescent mothers and their children in Ugu, KwaZulu-Natal. Afr J Prim Health Care Fam Med. 2019;11(1):1–11.
- 40. Zembe YZ, Townsend L, Thorson A, Silberschmidt M, Ekstrom AM. Intimate partner violence, relationship power inequity and the role of sexual and social risk factors in the production of violence among young women who have multiple sexual partners in a peri-urban setting in South Africa. PLoS ONE. 2015;10(11): e0139430.
- Groves AK, Gebrekristos LT, Reyes LM, Moodley D, Maman S. Describing relationship characteristics and postpartum HIV risk among adolescent, young adult, and adult women in South Africa. J Adolesc Health. 2020;67(1):123–6.
- 42. Hindin MJ, Kishor S, Ansara DL. Intimate Partner Violence among Couples in 10 DHS Countries: Predictors and Health Outcomes. Calverton, Maryland, USA: Macro International Inc; 2008. (DHS Analytical Studies). Report No.: No. 18.
- United Nations, Department of Economic and Social Affairs PD. World Urbanization Prospects: The 2022 Revision [Internet]. United Nations Publications New York; 2022. https://population.un.org/wpp/. Accessed 10 Sept 2022.
- 44. Haffejee F, Govender N, Reddy P, Sibiya MN, Ghuman S, Ngxongo T, et al. Factors associated with unintended pregnancy among women attending a public health facility in KwaZulu-Natal. South Africa South African Family Practice. 2018;60(3):79–83.
- Groves AK, Gebrekristos LG, Reyes HLM, Moodley D, Raziano V, Maman S. A mixed-methods study of resilience and return to school among adolescent mothers in South Africa. Glob Public Health. 2021. https://doi. org/10.1080/17441692.2021.1970208.
- Gebrekristos L, Groves AK, Shazi Z, Moodley D. Soliciting parental consent amongst adolescent minor mothers: a barrier in adolescent HIV research? S Afr Med J. 2021;111(6):526–7.

- Cox JL, Holden JM, Sagovsky R. Detection of Postnatal Depression. Br J Psychiatry. 1987;150(06):782–6.
- Kassa GM, Arowojolu AO, Odukogbe ATA, Yalew AW. Adverse maternal outcomes of adolescent pregnancy in Northwest Ethiopia: a prospective cohort study. PLoS ONE. 2021;16(9): e0257485.
- 49. World Health Organization. WHO multi-country study on women's health and domestic violence against women: summary report of initial results on prevalence, health outcomes and women's responses. Geneva: World Health Organization; 2005.
- Birkeland R, Thompson JK, Phares V. Adolescent motherhood and postpartum depression. J Clin Child Adolesc Psychol. 2005;34(2):292–300.
- Harrykissoon SD, Rickert VI, Wiemann CM. Prevalence and patterns of intimate partner violence among adolescent mothers during the postpartum period. Arch Pediatr Adolesc Med. 2002;156(4):325–30.
- Groves AK, Moodley D, McNaughton-Reyes L, Martin SL, Foshee V, Maman S. Prevalence, rates and correlates of intimate partner violence among South African women during pregnancy and the postpartum period. Matern Child Health J. 2015;19(3):487–95.
- 53. Mthembu J, Mabaso M, Reis S, Zuma K, Zungu N. Prevalence and factors associated with intimate partner violence among the adolescent girls and young women in South Africa: findings the 2017 population based cross-sectional survey. BMC Public Health. 2021;21(1):1160.
- Durevall D, Lindskog A. Intimate partner violence and HIV in ten sub-Saharan African countries: what do the Demographic and Health Surveys tell us? Lancet Glob Health. 2015;3(1):e34-43.
- Campbell JC, Baty ML, Ghandour RM, Stockman JK, Francisco L, Wagman J. The intersection of intimate partner violence against women and HIV/ AIDS: a review. Int J Inj Contr Saf Promot. 2008;15(4):221–31.
- Mokhele I, Nattey C, Jinga N, Mongwenyana C, Fox MP, Onoya D. Prevalence and predictors of postpartum depression by HIV status and timing of HIV diagnosis in Gauteng, South Africa. PLoS ONE. 2019;14(4): e0214849.
- Pitpitan EV, Kalichman SC, Eaton LA, Cain D, Sikkema KJ, Skinner D, et al. Gender-based violence, alcohol use, and sexual risk among female patrons of drinking venues in Cape Town, South Africa. J Behav Med. 2013;36(3):295–304.
- Peltzer K, Shikwane ME. Prevalence of postnatal depression and associated factors among HIV-positive women in primary care in Nkangala district, South Africa. South Afr J HIV Med. 2011;12(4):24–8.
- Sigalla GN, Rasch V, Gammeltoft T, Meyrowitsch DW, Rogathi J, Manongi R, et al. Social support and intimate partner violence during pregnancy among women attending antenatal care in Moshi Municipality, Northern Tanzania. BMC Public Health. 2017;17(1):240.
- Hassan M, Kashanian M, Roohi M, Yousefi H. Maternal outcomes of intimate partner violence during pregnancy: study in Iran. Public Health. 2014;128(5):410–5.
- Suhitharan T, Pham TPT, Chen H, Assam PN, Sultana R, Han NLR, et al. Investigating analgesic and psychological factors associated with risk of postpartum depression development: a case–control study. NDT. 2016. https://doi.org/10.2147/NDT.S105918.
- Filmer D, Pritchett LH. Estimating wealth effects without expenditure data-or tears: an application to educational enrollments in states of India. Demography. 2001;38(1):115–32.
- Van Der Berg S, Louw M. Changing patterns of South African income distribution: towards time series estimates of distribution and poverty. S Afr J Econ. 2004;72(3):546–72.
- 64. National Department of Health, South Africa. National consolidated guidelines for the prevention of mother-to-child transmission of HIV (PMTCT) and the management of HIV in children, adolescents and adults. South Africa: Republic of South Africa, Health Department; 2015.
- 65. Zou G. A modified poisson regression approach to prospective studies with binary data. Am J Epidemiol. 2004;159(7):702–6.
- Zou GY, Donner A. Extension of the modified Poisson regression model to prospective studies with correlated binary data. Stat Methods Med Res. 2013;22(6):661–70.
- 67. SAS Institute Inc. SAS version 9.4. Cary: SAS Institute Inc; 2013.
- Roberts KJ, Smith C, Cluver L, Toska E, Zhou S, Boyes M, et al. Adolescent motherhood and HIV in South Africa: examining prevalence of common mental disorder. AIDS Behav. 2022;26(4):1197–210.

- Mbawa M, Vidmar J, Chingwaru C, Chingwaru W. Understanding postpartum depression in adolescent mothers in Mashonaland Central and Bulawayo provinces of Zimbabwe. Asian J Psychiatr. 2018;32:147–50.
- Barhafumwa B, Dietrich J, Closson K, Samji H, Cescon-Colizza A, Nkala B, et al. High prevalence of depression symptomology among adolescents in Soweto, South Africa associated with being female and cofactors relating to HIV transmission. Vulnerable Child Youth Stud. 2016;22(11):1–11.
- Ajaero CK, Nzeadibe CT, Igboeli EE. Rural-urban differences in the prevalence and predictors of depression among adolescents in South Africa. S Afr J CH. 2018;12(2b):71.
- Pettifor A, MacPhail C, Selin A, Gómez-Olivé FX, Rosenberg M, Wagner RG, et al. HPTN 068: a randomized control trial of a conditional cash transfer to reduce HIV infection in young women in South Africa—study design and baseline results. AIDS Behav. 2016;20(9):1863–82.
- Cluver L, Orkin M, Boyes ME, Sherr L. Child and adolescent suicide attempts, suicidal behavior, and adverse childhood experiences in south africa: a prospective study. J Adolesc Health. 2015;57(1):52–9.
- Cooper PJ, Tomlinson M, Swartz L, Woolgar M, Murray L, Molteno C. Postpartum depression and the mother-infant relationship in a South African peri-urban settlement. Br J Psychiatry. 1999;175(6):554–8.
- Dadi AF, Akalu TY, Baraki AG, Wolde HF. Epidemiology of postnatal depression and its associated factors in Africa: a systematic review and meta-analysis. PLoS ONE. 2020;15(4): e0231940.
- Rogathi JJ, Manongi R, Mushi D, Rasch V, Sigalla GN, Gammeltoft T, et al. Postpartum depression among women who have experienced intimate partner violence: a prospective cohort study at Moshi, Tanzania. J Affect Disord. 2017;15(218):238–45.
- World Health Organization. WHO recommendations on maternal and newborn care for a positive postnatal experience [Internet]. Geneva, Switzerland; 2022. https://www.who.int/publications-detail-redirect/ 9789240045989. Accessed 18 Sep 2022.
- World Health Organization. WHO recommendations on antenatal care for a positive pregnancy experience. Geneva: World Health Organization; 2016.
- 79. Bhusal BR, Bhandari N, Chapagai M, Gavidia T. Validating the Edinburgh Postnatal Depression Scale as a screening tool for postpartum depression in Kathmandu. Nepal Int J Ment Health Syst. 2016;10(1):71.
- Garcia-Esteve L, Ascaso C, Ojuel J, Navarro P. Validation of the Edinburgh Postnatal Depression Scale (EPDS) in Spanish mothers. J Affect Disord. 2003;75(1):71–6.
- Reddy P, Sewpaul R, Jonas K. Teenage pregnancy in South Africa: reducing prevalence and lowering maternal mortality rates. Human Sciences Research Council; 2016. HRSC Policy Brief. https://repository.hsrc.ac.za/ handle/20.500.11910/9925. Accessed 13 Mar 2023.
- 82. Yah CS, Ndlovu S, Kutywayo A, Naidoo N, Mahuma T, Mullick S. The prevalence of pregnancy among adolescent girls and young women across the Southern African development community economic hub: a systematic review and meta-analysis. Health Promot Perspect. 2020;10(4):325.

### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

